Author's response to reviews

Title: Perceptions of frontline staff regarding data collection methodologies used during the 2009 A H1N1 influenza immunization campaign in Canada

Authors:

Julie Foisy (julie.foisy@oahpp.ca)
Susan Quach (susan.quach@oahpp.ca)
Christine L Heidebrecht (christine.heidebrecht@oahpp.ca)
Jennifer A Pereira (jennifer.pereira@oahpp.ca)
Sherman D Quan (Sherman.Quan@uhn.on.ca)
Maryse Guay (Maryse.Guay@USherbrooke.ca)
Julie A Bettinger (jbettinger@cfri.ca)
Shelley L Deeks (shelley.deeks@oahpp.ca)
Stephanie Brien (stephanie.brien@mail.mcgill.ca)
Jeffrey C Kwong (jeff.kwong@utoronto.ca)

Version: 5 Date: 23 December 2010

Author's response to reviews: see over
December 23, 2010

Dr. Melissa Norton, Editor-in-Chief
BMC Public Health
BioMed Central Ltd
Middlesex House
34-42 Cleveland Street
London UK
W1T 4LB

Dear Dr. Norton:

On behalf of the Public Health Agency of Canada/Canadian Institutes of Health Research Influenza Research Network (PCIRN) Vaccine Coverage Theme Group, I am pleased to submit the final revised version of the manuscript, “Perceptions of frontline staff regarding data collection methodologies used during the 2009 A H1N1 influenza immunization campaign in Canada”, for publication in the BMC Public Health Journal.

The editorial production team’s comments have been pasted into the table below along with the authors’ reply. There is one request I want to highlight from the table: would it be possible to add “for the Public Health Agency of Canada/Canadian Institutes of Health Research Influenza Research Network (PCIRN) Vaccine Coverage Theme Group” after the last author, in the author list? I could not add the group as an author in the submission process.

Happy Holidays!

Sincerely,

Julie Foisy
<table>
<thead>
<tr>
<th>Editorial Team Comments</th>
<th>Author's Reply</th>
</tr>
</thead>
<tbody>
<tr>
<td>Study Groups: these should be consistent between the manuscript and the submission system. Members of a group should be listed in the acknowledgements (if anywhere).</td>
<td>The PCIRN team members are listed in the acknowledgement section; however please include “for the Public Health Agency of Canada/Canadian Institutes of Health Research Influenza Research Network (PCIRN) Vaccine Coverage Theme Group” after the last author in the author list. I could not add the group as an author in the submission process.</td>
</tr>
<tr>
<td>Table 2: tables must be created using a Tables tool, and should not be linked files, figures, or generated with tabbed text.</td>
<td>Done.</td>
</tr>
<tr>
<td>Additional files: Please ensure all files are referred to as additional files (not, for example, supplementary data).</td>
<td>Done.</td>
</tr>
<tr>
<td>Additional files: please add a section titled &quot;Additional files&quot; at the end of the manuscript (after the tables) listing the following for each file: the title of the data, and a short description of the data. An example is given below:</td>
<td>Done.</td>
</tr>
<tr>
<td>Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.</td>
<td>Done.</td>
</tr>
<tr>
<td>References: please provide the full citation details for references 3, 7 and 9.</td>
<td>Done.</td>
</tr>
<tr>
<td>Tables- please remove the visible vertical lines from your tables.</td>
<td>Done.</td>
</tr>
<tr>
<td>Tables: we are unable to display shading and colours within the tables. Please remove the shading from table 1.</td>
<td>Done.</td>
</tr>
<tr>
<td>Appendix: please remove the appendix from the manuscript as this has already been included as an additional file.</td>
<td>Done.</td>
</tr>
<tr>
<td>Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.</td>
<td>Done.</td>
</tr>
</tbody>
</table>